Author's response to reviews

Title: The Effects of Kenya's Emergency Hire Nursing Program on the Delivery of Health Services: a Pilot Evaluation within Two Districts

Authors:

Stephen M Vindigni (stephen.vindigni@gmail.com)
Patricia L Riley (pyr0@cdc.gov)
Francis Kimani (dmskenya@gmail.com)
Rankesh Willy (rankwill2002@yahoo.com)
Patrick Warutere (teshpatt@yahoo.com)
Jennifer F Sabatier (fvv9@cdc.gov)
Rose Kiriinya (7.rose.kiriinya@emorykenya.org)
Michael Friedman (mff7@cdc.gov)
Martin Osuma (mosumba@yahoo.ca)
Agnes N Waudo (agnes.waudo@emorykenya.org)
Chris Rakuom (cprakuom2009@rocketmail.com)
Martha F Rogers (mrogers@taskforce.org)

Version: 3
Date: 25 February 2014

Author's response to reviews:

In response to the Editors' comments:

1. Please provide Competing interests

The following text has been added to the manuscript:

All authors declare: no support from any organization for the submitted work; no financial relationships with any organizations that might have an interest in the submitted work. The authors declare that they have no competing interests.

2. Please include Authors' contributions.

This has been included at the end of the document after acknowledgements.

3. Figure legends

There are 2 figures in this manuscript. Figure 1: the title has been removed from the graphic and included in the text. Since the image is color, the legend is part of the document. Figure 2 contains several graphics, each with their own color legend.

The authors are more than willing to discuss additional modifications to the graphics/legends to best fit with the Journal's format.